Author's response to reviews

Title: Cardiovascular disease prevention in rural Nigeria in the context of a community based health insurance scheme: QUality Improvement Cardiovascular care Kwara-I (QUICK-I)

Authors:

Marleen Hendriks (m.hendriks@amc-cpcd.org)
Lizzy Brewster (L.M.Brewster@amc.uva.nl)
Ferdinand Wit (f.wit@amc-cpcd.org)
Oladimeji Akeem Bolarinwa (drdeji@yahoo.com)
Aina Olufemi Odusola (A.O.Odusola@amc.uva.nl)
William Redekop (redekop@bmg.eur.nl)
Navin Bindraban (n.r.bindraban@amc.uva.nl)
Albert Vollaard (a.vollaard@pharmaccess.org)
Shade Alli (sadealling@yahoo.com)
Peju Adenusi (dradenusi@hygeiagroup.com)
Kayode Agbede (m.hendriks@amc-cpcd.org)
Tanimola Akande (akandetm@yahoo.com)
Joep Lange (j.lange@amc.uva.nl)
Constance Schultsz (schultsz@gmail.com)

Version: 3 Date: 13 March 2011

Author's response to reviews:

Dear editor,

In response to the email of the BMC Series Editorial Production Team dated 8-3-2011 we have made the following changes to the format of manuscript MS: 1281588731518325.

The affiliations were changed such that they match the format Department, Institute, (Street), City, (Postal code), Country. Some of the Nigerian addresses do not have streets or postal codes. We have used the official addresses of the institutions.

The Methods subheading in the abstract was changed to Methods/Design

The figure was cropped as much as possible

Vertical lines from the tables were removed

The document was checked for typography

We hope these changes are consistent with the BMC format, please let us know if any additional changes are required.

Kind regards,